Author’s response to reviews

Title: Pessaries in multiple pregnancy as a prevention of preterm birth: the ProTwin Trial

Authors:

Maud A Hegeman (protwintrial@gmail.com)
Dick J Bekedam (d.j.bekedam@olvg.nl)
Kitty WM Bloemenkamp (k.w.m.bloemenkamp@lumc.nl)
Anneke Kwee (a.kwee@umcutrecht.nl)
Dimitri NM Papatsonis (hoog.pap@wxs.nl)
Joris AM van der Post (j.a.vanderpost@amc.uva.nl)
Arianne C Lim (a.c.lim@amc.uva.nl)
Hubertina CJ Scheepers (hcj.scheepers@mumc.nl)
Christine Willekes (cwi@sgyn.azm.nl)
Johannes J Duvekot (j.j.duvekot@erasmusmc.nl)
Marc Spaanderman (M.Spaanderman@obgyn.umcn.nl)
Martina Porath (m.porath@mmc.nl)
Jim van Eyck (jvaneyck@wxs.nl)
Monique C Haak (mc.haak@vumc.nl)
Maria G van Pampus (m.van.pampus@og.umcg.nl)
Hein W Bruinse (H.W.Bruinse@umcutrecht.nl)
Ben Willem J Mol (b.w.mol@amc.uva.nl)

Version: 5 Date: 8 September 2009

Author’s response to reviews: see over
Subject: Revised Manuscript “Pessaries in multiple pregnancy as a prevention of preterm birth: the ProTwin Trial” (MS: 8966035222867889)

Dear Editor,

As you requested in your email on 28th of August 2009, we revised the manuscript “Pessaries in multiple pregnancy as a prevention of preterm birth: the ProTwin Trial” on the following points:

- We removed the word “ProTwin study group” from the title page.
- We added a note describing the role of the group in the Authors’ contributions section.

We hope you will accept this study protocol for online publication in BMC Pregnancy and Childbirth.

Yours sincerely,

Maud Hegeman, MD, corresponding author

Onze Lieve Vrouwe Gasthuis
Department of Obstetrics and Gynaecology
Oosterpark 9,
1091 AC Amsterdam
The Netherlands
Tel: +31 6 14866288
Email: protwintrial@gmail.com